Author's response to reviews

Title: Termite usage associated with antibiotic therapy: Enhancement of aminoglycoside antibiotic activity by natural products of Nasutitermes corniger (Motschulsky 1855)

Authors:

Henrique DM Coutinho (hdmcoutinho@gmail.com)
Alexandre Vasconcellos (alexyvasconcellos@yahoo.com.br)
Micheline A Lima (michelinealima@hotmail.com)
Rômulo RN Alves (romulo_nobrega@yahoo.com.br)
Geraldo G Almeida-Filho (citopat@hotmail.com)

Version: 4 Date: 16 September 2009

Author's response to reviews: see over
Response to the editorial review of the article MS: 1798945572655256 - Termite usage associated with antibiotic therapy: enhancement of aminoglycoside antibiotic activity by natural products of Nasutitermes corniger (Motschulsky 1855).

Dear editor, I would like to thanks you and the reviewers by the valious suggestions and critics. About this ones, I would like to expose for you these follow considerations:

Editorial Reviewer's report

Title: Termite usage associated with antibiotic therapy: Enhancement of aminoglycoside antibiotic activity by natural products of Nasutitermes corniger (Motschulsky 1855)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

DONE

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns). For example you would change 'The Title of My Prestigious Publication: A Scientific Study¿. To 'The title of my prestigious publication: a scientific study¿.

DONE

Please place the competing interests before the Authors' contributions.

DONE

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

DONE

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

PAPER REVISED

Again, I would like to thank you by the suggestions and critics.

Kind regards.

HENRIQUE DOUGLAS MELO COUTINHO